Author's response to reviews

Title: Structural equation modeling of factors contributing to quality of life in Japanese patients with multiple sclerosis

Authors:

Hiromi Kikuchi (h.kikuchi@scu.ac.jp)
Nobuhiro Mifune (n.mifune@gmail.com)
Masaaki Niino (niino@hok-mc.hosp.go.jp)
Jun-ichi Kira (kira@neuro.med.kyushu-u.ac.jp)
Tatsu Koriyama (koriyama@city-hosp.naka.hiroshima.jp)
Kohei Ota (kota@rs.kagu.tus.ac.jp)
Masami Tanaka (nra17777@nifty.com)
Hirofumi Ochi (ochi@m.ehime-u.ac.jp)
Shunya Nakane (snakane@nkmc1.jp)
Seiji Kikuchi (skikuti@hmc.hosp.go.jp)

Version: 2 Date: 7 October 2012

Author's response to reviews:

Dr. Deesha Majithia
Executive Editor of BMC Neurology

Dear Dr. Majithia,

We would like to re-submit our manuscript entitled “Structural equation modeling of factors contributing to quality of life in Japanese patients with multiple sclerosis” to Research Article of BMC Neurology.

Details of Questionnaire items were included in the manuscript as a Table and some comments were added according to your suggestions.

We believe that the data described in this manuscript will be of special interest to the readers of BMC Neurology.

Sincerely,

Masaaki Niino, MD, PhD
Director, Department of Clinical Research
Hokkaido Medical Center
Yamanote 5-jo 7-chome, Nishi-ku,
Sapporo 063-0005 JAPAN
Tel +81-11-611-8111, Fax +81-11-611-5820
email: niino@hoc-mc.hosp.go.jp